Author’s response to reviews

Title: Patients with chronic Pain: Evaluating Depression and their Quality of Life in a single center study in Greece

Authors:

Ekaterini Rapti (ksprapti2@gmail.com)
Dimitrios Damigos (ddamigos@gmail.com)
Paraskevi Apostolara (v.apostolara@gmail.com)
Vasiliki Roka (vassoroka2003@yahoo.gr)
Christos Lionis (lionis@galinos.med.uoc.gr)

Version: 5 Date: 20 Nov 2019

Author’s response to reviews:

1. While assessing your manuscript it was noticed that Chara Tzavara was not accredited in the Authors contributions’ sub-section of the Declarations. Please be sure to include all individuals accredited with authorship in this section along with their individual contributions to the study and/or manuscript.

   Answer: Done

2. Please include a statement in Authors’ contributions clearly indicating that all authors read and approved the final version of the manuscript.

   Answer: Done

3. Please move the List of Abbreviations section to before the Declarations section.

   Answer: Done

4. Please remove the author response to reviewers.

   Answer: Done
5. Please remove the duplicate tables that are attached in the manuscript.

   Answer: Done

6. Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box onl

   Answer: Done